Author’s response to reviews

Title: AZGP1 inhibits soft tissue sarcoma cells invasion and migration

Authors:

Jiayong Liu (liuijayong_doc@163.com)
Haibo Han (hanhaibohyd@sina.com)
Zhengfu Fan (fanzhengfu@126.com)
Marc El Beaino (marcbeaino@gmail.com)
Zhiwei Fang (fzw66@aliyun.com)
Shu Li (lishu_doctor@163.com)
Jiafu Ji (jijiafu@hsc.pku.edu.cn)

Version: 4 Date: 24 Oct 2017

Author’s response to reviews:

Dear Dr. Strong,

Thank you very much for your patience and careful review. For the authorship of the manuscript, we checked the guidelines carefully again, discussed with all the authors and made a decision as below:

Dr. ZFF did a lot of work for patients’ follow-up and acquisition of clinical data, so we kept him as a coauthor of the manuscript. We removed Drs. CJB and RFX as they don’t meet the qualification for authorship.

So we have made some changes in the manuscript according to our decision: Drs. CJB, RFX and their information are removed from the list of authors.

In the section of ‘Authors contributions’, Drs. CJB, RFX are removed, and a passage ‘Acquisition of clinical data and follow-up of patients’ is added for contribution of ZFF. 416-417, page 24.

I greatly appreciate all your time and effort. Please do not hesitate to contact me if you have any questions or concerns.

Best regards!
Sincerely,

Jiafu Ji